Author's response to reviews

Title: GSTM1 and APE1 genotypes affect arsenic-induced oxidative stress: a repeated measures study

Authors:

Carrie V Breton (breton@usc.edu)
Molly L Kile (mkile@hsph.harvard.edu)
Paul J Catalano (pcata@jimmy.harvard.edu)
Elaine Hoffman (ehoffman@hsph.harvard.edu)
Quazi Quamruzzaman (dch@bangla.net)
Mahmuder Rahman (dch@bangla.net)
Golam Mahiuddin (dch@bangla.net)
David C Christiani (dchris@hsph.harvard.edu)

Version: 4 Date: 21 November 2007

Author's response to reviews: see over
Dear Dr. Ozonoff,

I have made the following edits, per your request.

1. In the title page I have included country (USA), and added email addresses of all authors.
2. 'Conclusion' was pluralized.
3. In references, I made sure all authors were listed and no 'et al' was used.
5. For tables, I made sure the borders of each cell display as black lines.